

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Black, Lamont K.; Rosen, Richard K.

## Working Paper How the credit channel works: Differentiating the bank lending channel and the balance sheet channel

Working Paper, No. 2007-13

**Provided in Cooperation with:** Federal Reserve Bank of Chicago

*Suggested Citation:* Black, Lamont K.; Rosen, Richard K. (2007) : How the credit channel works: Differentiating the bank lending channel and the balance sheet channel, Working Paper, No. 2007-13, Federal Reserve Bank of Chicago, Chicago, IL

This Version is available at: https://hdl.handle.net/10419/70522

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



# WWW.ECONSTOR.EU



Federal Reserve Bank of Chicago

## How the Credit Channel Works: Differentiating the Bank Lending Channel and the Balance Sheet Channel

Lamont K. Black and Richard J. Rosen

WP 2007-13

### How the Credit Channel Works: Differentiating the Bank Lending Channel and the Balance Sheet Channel

by

Lamont K. Black Board of Governors of the Federal Reserve System Washington, DC 20551 U.S.A

and

Richard J. Rosen Federal Reserve Bank of Chicago Chicago, IL 60604 U.S.A

This draft: October 2007

#### ABSTRACT:

The credit channel of monetary policy transmission operates through changes in lending. To examine this channel, we explore how movements in the real federal funds rate affect bank lending. Using data on individual loans from the Survey of Terms of Bank Lending, we are able to differentiate two ways the credit channel can work: by affecting overall bank lending (the bank lending channel) and by affecting the allocation of loans (the balance sheet channel). We find evidence consistent with the operation of both internal credit channels. During periods of tight monetary policy, banks adjust their stock of loans by reducing the maturity of loan originations and they reallocate their short-term loan supply from small firms to large firms. These results are stronger for large banks than for small banks.

The opinions expressed do not necessarily reflect those of the Federal Reserve Board, the Federal Reserve Bank of Chicago, or their staffs. We would like to thank Shah Hussain for research assistance, John Duca for comments, and participants in presentations at the Federal Reserve Bank of Chicago, the Federal Reserve Bank of San Francisco, and the Federal Reserve System Conference on Bank Structure and Competition.

#### 1. Introduction

Understanding the effect of monetary policy on bank lending is important for understanding the transmission mechanism of monetary policy. Models of the credit channel show that financial frictions can magnify the effects of monetary policy. Within the credit channel, there are two channels by which these financial frictions can operate: the bank lending channel and the balance sheet channel. Our main contribution is to empirically differentiate these two channels, identifying the specific contribution of each channel within bank lending, something not generally done in previous studies. We find evidence that during periods of tight monetary policy, banks adjust their stock of loans by reducing the maturity of loan originations and they reallocate their short-term loan supply from small firms to large firms. These results are consistent with the operation of both internal credit channels. We show that the bank lending channel operates through changes in loan maturity and the balance sheet channel operates through a reallocation of short-term loan supply.

The division of the credit channel into the bank lending channel and the balance sheet channel is made in Bernanke and Gertler (1995). In each channel, the effect of monetary policy is magnified due to a change in lending. The theory of the bank lending channel is based on the Bernanke and Blinder (1988) model, where a contractionary monetary policy leads to a reduction in bank deposits which, in turn, reduces aggregate loan supply. An alternative channel is described in Bernanke, Gertler, and Gilchrist (1996). There, the agency costs of lending endogenously change with monetary policy. Monetary contractions reduce the net worth of borrowers, which leads to an increase in agency costs, primarily for low-networth firms. In the Bernanke and Gertler (1989) model, when these agency costs increase, lenders reduce the amount of credit extended to risky firms and invest more in a safe alternative. This leads to our interpretation that the balance sheet channel implies that during periods of tight monetary policy, banks reallocate loan supply away from small firms and toward large firms, with the implicit assumption than small firms are riskier than large firms.

There is a significant literature on both the bank lending channel and the balance sheet channel. The evidence for the bank lending channel is based primarily on differences among banks (e.g. Kashyap and Stein (1995, 2000); Kishan and Opiela (2000); Jayaratne and Morgan (2000); Ashcraft (2006)). This literature generally shows that banks with weak

balance sheets (such as limited liquidity, limited capitalization, etc.) reduce their loan supply following a monetary contraction. The evidence for the balance sheet channel is based primarily on differences among firms. Cross-sectional analysis of firms shows that, following a monetary contraction, small firm debt decreases and large firm debt increases (Oliner and Rudebusch (1996); Bernanke, Gertler, and Gilchrist (1996)). A conclusion that one channel is operating is often asserted without differentiating the specific contributions of the two channels. Some papers explicitly state that they identify a credit channel, but cannot empirically separate the bank lending channel from the balance sheet channel (e.g. Kashyap, Lamont, and Stein (1994); Gertler and Gilchrist (1994); Morgan (1998)).

Our main contribution is to differentiate how the two channels operate. By using loanlevel data from the Survey of Terms of Business Lending (STBL), we can separate changes in loan supply from changes in loan demand. This allows us to examine whether banks are changing aggregate supply and, within supply, changing the type of loans they make (and the borrowers they make them to). It is important to distinguish the two channels, because the effect of monetary policy on credit availability can differ for some borrowers across the two channels.

The key insight in our identification strategy is that banks make many of their loans under pre-existing loan commitments. It is less likely that a bank can alter the terms of a commitment loan than the terms of a loan not made under commitment (that is, a spot loan). When the level of monetary tightness changes, banks may want to respond by changing the level of their loan supply. The supply of spot loans can be changed instantaneously, because banks can immediately adjust their loan approval criteria. Banks can tighten their lending standards, for instance, and immediately make it more difficult for lower quality firms to acquire a loan. The supply of commitment loans, on the other hand, cannot be changed as readily. Firms with a pre-existing commitment can continue to draw funding from their bank as long as the firm does not experience a material adverse change in their credit quality.<sup>1</sup> Therefore, the supply of commitment loans is stickier than the supply of spot loans.

The restriction on adjustments to commitment loan supply allows us to identify changes in loan supply apart from changes in loan demand. If demand is relatively consistent

<sup>&</sup>lt;sup>1</sup> Some firms may enter into new commitments right before drawing them down, but we provide evidence that most commitment drawdowns are made under commitments signed at least 6 months prior to the drawdown.

across the two types of borrowing, we can identify changes in the supply of spot loans by comparing spot lending *relative* to commitment lending. We assume that changes in the level of commitment lending reflect changes in loan demand. These could occur through the traditional "interest-rate" channel by which monetary policy affects firms' cost of borrowing. So, if the level of spot lending and commitment lending both rise or fall in tandem when monetary policy is altered, we attribute this to changes in loan demand rather than supply. However, if spot lending changes relative to commitment lending, we attribute the difference to a movement in loan supply. The important argument here is that commitment lending provides a cross-sectional baseline against which to evaluate changes in spot lending. It is the changes in spot lending relative to commitment lending that provides information about the supply of spot loans.

This approach matches nicely to our analysis of firm size, because it can be applied to small firms and large firms alike. This differs from the approach of Kashyap, Stein, and Wilcox (1993) who use outstanding commercial paper as a baseline for loan demand. Using public debt as a baseline for firm demand is feasible for large firms, but not for small firms, because small firms do not issue commercial paper or other forms of public debt (i.e., bonds). Our approach allows us to separately analyze bank loan supply to small and large firms by using commitment loans as our baseline.

When grouping all loans together and treating all loan originations equally, we do not find evidence of a decrease in aggregate loan supply as monetary policy tightens. Some results even indicate a slight increase in loan supply. While this is not supportive of the bank lending channel, the results can be misleading because they group different types of loans together. In order to more precisely test whether the bank lending channel operates, and to test whether the balance sheet channel also operates, we divide the data in several ways.

The STBL measures the *flow* of lending by a bank while the bank lending channel makes predictions about the *stock* of loans. If a bank shifts from making long-term loans to short-term loans, fairly quickly its stock of loans can decrease even if the flow of loans remains constant or increases. We find that, during periods of tight credit, banks reallocate their loan supply from long-term lending to short-term lending, consistent with the operation of a bank lending channel.

We then examine a proxy for firm size to test the balance sheet channel. The STBL does not have borrower information, so we use loan size as a proxy for firm size. We find that banks shift their lending from small firms to large firms, mostly due to changes in short-term lending. This suggests that banks may shift lending toward safe, transparent firms during periods of tight monetary policy, consistent with the balance sheet channel.

Our combined results help to differentiate the specific contributions of the two aspects of the credit channel. It appears that banks may reduce loan supply during periods of tight monetary policy by shortening loan maturities and thereby increasing the liquidity of their loan portfolio. It also appears that banks reallocate short-term lending from small to large firms, which may be a means of increasing the quality of their loan portfolio. These two effects separately identify how the bank lending channel and the balance sheet channel operate.

The remainder of the paper is organized as follows. Section 2 reviews the literature and section 3 describes the data used in our analysis. Section 4 explains our empirical methodology and section 5 lays out the results. Section 6 concludes.

#### 2. Literature Review

In the traditional view of how monetary policy works, changes in monetary policy affect the real economy through the interest-rate channel. According to this view, policymakers set short-term real interest rates to influence the cost of capital, which affects business investment and demand for consumer durables. These changes in aggregate demand then affect the level of production (see Bernanke and Blinder (1992)).

Changes in monetary policy can also affect the supply side of the credit market. This is known as the credit channel. Financial frictions stemming from information asymmetries may be related to the underlying cost of capital. If the cost of external financing increases with short-term real interest rates, then this can reduce the availability of credit during periods of tight monetary policy.

Within the credit channel, the bank lending channel and the balance sheet channel have been proposed as two distinct channels (Bernanke and Gertler (1995)). Each is unique in how it operates, yet the two are very similar in their empirical predictions. Consequently, much of the literature has either estimated the channels in isolation or explicitly treated them

as one channel. In this section, we discuss the evidence for each of the channels as well as the evidence for the credit channel as a whole. Our paper builds on this work. We differentiate the channels by providing evidence on how the two channels operate.

The bank lending channel predicts that banks reduce their aggregate supply of loans when monetary policy is tight. The process begins with a monetary contraction which drains reserves from the banking system and this leaves banks with fewer funds to lend (Bernanke and Blinder (1988)). It is possible that banks can react to this by increasing non-reservable liabilities such as uninsured deposits, thus leaving lending unaffected (Romer and Romer (1990)). However, since raising non-reservable liabilities is costly, tightening monetary policy is expected to still result in a net reduction in loanable funds (Stein (1998)).

A number of papers have used cross-sectional variation in bank characteristics to identify changes in loan supply and have found evidence for the bank lending channel. Kashyap and Stein (1995, 2000), for instance, use bank size and liquidity to differentiate banks. The argument is that the lending of small banks with illiquid balance sheets should be most sensitive to changes in monetary policy, because raising wholesale liabilities is more costly for them. The authors find that small banks with liquid balance sheets reduce their lending less in response to monetary contractions than other small banks. Other studies find similarly that the effects of monetary tightening are decreasing in the expected costs of raising non-reservable liabilities.<sup>2</sup>

The balance sheet channel predicts that banks reallocate their loan supply away from small firms and toward large firms when monetary policy is tight.<sup>3</sup> This channel has also been referred to as the "flight to quality." Lang and Nakamura (1995), for instance, show that banks make proportionately more "safe" loans during periods of tight credit. Because agency costs are greater for low-net-worth firms, large firms tend to be safer than small firms. Bernanke, Gertler, and Gilchrist (1996) show cross-sectional implications of the balance sheet channel by firm characteristics. The authors hypothesize that borrowers who face significant agency costs should experience reduced access to credit, relative to other borrowers,

<sup>&</sup>lt;sup>2</sup> Kishan and Opiela (2000) examine the effects of bank capitalization, Jayaratne and Morgan (2000) look at dependence on core deposits, and Ashcraft (2006) analyzes banks by holding company status.

<sup>&</sup>lt;sup>3</sup> The more general form of the balance sheet channel refers to a reduction in loan supply to low-quality borrowers relative to high-quality borrowers.

following a monetary contraction. Using data for manufacturing firms, Bernanke et al. find that small firms reduce their debt and large firms increase their debt.

It is important to clarify that the bank lending channel does not imply differences between changes in bank loan supply to large and small firms. The bank lending channel predicts that, following a monetary contraction, banks reduce their aggregate loan supply. Earlier studies take advantage of the fact that only some firms can substitute into non-bank sources of debt. The "bank-dependent" firms face a credit constraint due to the reduction in bank loan supply, but firms with available external finance can substitute into non-bank finance. The difference between firms comes from a difference in outside options. The theory makes no prediction about banks themselves reallocating their loan supply across firms. Therefore, large and small firms should face a similar reduction in loan supply, regardless of whether they have access to non-bank debt. In our data, we are looking at firm differences solely *within* bank loan supply, which is one of the advantages of our approach. The bank lending channel makes no prediction about such a difference.

Similar concepts to the credit channel can also be found in the credit rationing literature. Stiglitz and Weiss (1981) show how credit rationing can exist in equilibrium. If moral hazard increases with increased interest rates, banks may optimally price loans in a way that leads to rationing. In the benchmark Stiglitz-Weiss model, there is rationing of all borrowers (observationally indistinguishable) when rates are high, which is similar to the bank lending channel. In an extension of the model to include borrower groups, the authors show that lower quality groups only receive loans if credit is not rationed for higher quality groups. This is rationing of distinguishably low-quality borrowers, which is similar to the balance sheet channel. Berger and Udell (1992) use the STBL data to test for credit rationing at the aggregate level. However, the authors do not differentiate by firm size, so they are unable to differentiate between the bank lending channel and the balance sheet channel.

Some papers identify a credit channel, but do not distinguish the bank lending channel from the balance sheet channel. Kashyap, Lamont, and Stein (1994) analyze inventories for bank-dependent firms and find that the inventories of bank-dependent firms fall faster than the inventories of non-bank-dependent firms following a monetary contraction. Gertler and Gilchrist (1994) use a similar approach to analyze the inventories of small and large firms.<sup>4</sup> The authors find that the debt of small firms falls dramatically whereas large firm debt increases, which indicates changes in both loan supply and demand. Morgan (1998) uses aggregate spot loans and commitment loans and shows that spot loans fall relative to commitment loans following a contraction. The difficulty with using the borrower side is that it is difficult to separate the two channels within the credit channel. For instance, Gertler and Gilchrist do not attempt to separate the channels, because "the sets of borrowers who are balance-sheet constrained and who are bank-dependent overlap considerably" (p312).

One of the most influential approaches toward identifying changes in loan supply is based on the amount of non-bank finance used by firms. Kashyap, Stein, and Wilcox (1993) (hereinafter KSW) analyze the mix of bank debt and commercial paper (CP) debt to identify the bank lending channel. The authors find that the mix of bank debt relative to commercial paper debt decreases following a monetary contraction, which they interpret to be evidence for the bank lending channel.

The external finance approach to identification has raised questions about the nature of the credit channel. Oliner and Rudebusch (1996) argue that Kashyap et al. are measuring a shift in funding from small firms to large firms, which they call the broad credit channel. Oliner and Rudebusch make their argument by using the KSW approach to look at the debt mix of small and large firms. They note that small firms do not issue commercial paper, so they include "other" non-bank, non-CP debt in their measure of firms' baseline borrowing. KSW reply by questioning both the logic of the critique and this alternative mix measure for small firms which includes "other debt".

The value of the KSW approach is that commercial paper debt provides a baseline for measuring changes in the supply of bank debt to large firms. The early work on the credit channel focused on the response of aggregate lending to a monetary contraction (e.g. King (1986); Bernanke and Blinder (1992)), but did not distinguish between changes in loan supply and loan demand. The limitation of the KSW approach is that non-bank/non-CP debt is not a clear substitute for bank debt when the borrower is a small firm. Therefore, identifying changes in loan supply for both large firms and small firms requires a different approach.

<sup>&</sup>lt;sup>4</sup> Gertler and Gilchrist (1994) use changes in sales, inventories, and short-term debt to identify liquidity constraints. They cannot address the two parts of the credit channel, because their firm data does not differentiate the mechanisms of the two channels.

Our approach is similar to the KSW approach, but we use a different cross-sectional baseline for identifying changes in loan supply. We use the share of spot lending in total lending as our baseline and we argue that a reduction in spot lending relative to commitment lending indicates a reduction in loan supply. This is similar to the argument that a reduction in bank debt relative to commercial paper indicates a reduction in loan supply. The advantage of our approach is that we can address changes in both the supply of bank loans to small firms and the supply of bank loans to large firms.<sup>5</sup> This allows us to address the relative importance of the bank lending channel and the balance sheet channel. Our results provide additional evidence in support of the claim that the balance sheet channel has a greater effect than the bank lending channel.

#### 3. Data and summary statistics

The data are from the Survey of Terms of Business Lending (STBL), which is a survey conducted by the Federal Reserve Board of Governors. The STBL asks a sample of banks to provide data on every C&I loan issued during the first full business week of the second month of every quarter.<sup>6</sup> Our sample includes observations from the third quarter of 1982 through the first quarter of 2006, a time frame that covers numerous periods of tight credit.<sup>7</sup> Each quarter, the STBL includes a stratified sample of roughly 340 banks, including both small banks and large banks.<sup>8</sup> While banks move in and out of the STBL panel, the median length of time a bank is surveyed during our sample is 68 quarters.

Our empirical methodology analyzes the data on a loan-by-loan basis to test how loan supply changes with changes in monetary policy. This approach differs from Berger and Udell (1992), who construct aggregate observations for each bank in each quarter. To reduce noise in the data's time-series dimension, we apply additional filters to remove banks which

<sup>&</sup>lt;sup>5</sup> Commitments have been used, but only at the aggregate level (e.g., Berger and Udell (1992) and Morgan (1998)).

<sup>&</sup>lt;sup>6</sup> Although the survey includes several loan types, we focus strictly on commercial and industrial (C&I) loans in the survey (95.0% of the STBL loans are C&I loans). We also eliminate several groups of loans which are not appropriate to our analysis. We exclude all add-on loans, loans booked at a foreign office, loans with maturity greater than 10 years, and loans for which the interest rate spread over the Treasury of comparable maturity is less than -1 percent or greater than 10 percent.

<sup>&</sup>lt;sup>7</sup> The STBL begins in 1977, but we exclude observations prior to the third quarter of 1982 because there is no data on the size of commitment lines.

<sup>&</sup>lt;sup>8</sup> Some large banks report fewer than five days and include only a subset of total branches. We account for this when we do our analysis.

have a limited relevance to our analysis or limited presence in the STBL survey. First, we remove banks which issue almost all their loans under commitment or almost all their loans as spot loans, since these banks are unlikely to respond to changes in monetary policy by changing the mix of spot and commitment loans. Formally, we drop all banks where the average share of commitment loans in total loans is above 99% or below 1%. Second, we remove banks whose are in the survey fewer than 25 times. These banks have a limited time span in the survey, which reduces the possibility for changes over time. Finally, we remove banks which make less than 25 loans per week in the survey on average. With these filters, the final sample is 215 banks and 2,588,097 loans.

Table 1 lists the variables used in our tests along with the descriptive statistics of means and standard deviations. The loan characteristic variables are derived directly from the STBL data and the bank characteristics come from the Consolidated Report of Condition and Income (a.k.a. Call Report) associated with the bank making the loan. The STBL does not provide any information about the characteristics of the borrowing firms; therefore, all firm characteristics are imputed from loan level contract information. All nominal amounts are reported in 2006 US dollars.

The primary loan characteristic we use in our identification strategy is whether a loan is made under commitment. A commitment is either a formal commitment or an informal credit line which provides a borrower with the right to borrow up to a certain amount of credit over a fixed period of time. The important aspect of a commitment is that a bank is committed to making a loan upon request, independent of market conditions. Only a legal condition such as a clause for material adverse change (MAC) could release the bank from its commitment. Let COMMIT be a dummy variable equal to 1 if a loan is made under commitment and equal to 0 if the loan is a spot loan, which we define as a loan made to a firm when there was no prior commitment. The mean value of COMMIT is 0.816, which means that 81.6% of the STBL loans from 1982:Q3 to 2006:Q1 are made under commitment.

Our primary macroeconomic variable is our measure of monetary policy, which is the real federal funds rate (referred to as FUNDS RATE or "the funds rate"). The theory of the credit channel relates loan supply to real interest rates; therefore, we adjust the nominal funds rate for expected inflation. Our main measure of expected inflation is ex-ante inflation, so, in

other words, we use current inflation as the forecast of the next quarter's inflation. We construct the real federal funds rate as

$$FUNDS RATE = federal funds rate - current inflation,$$
(1)

where the federal funds rate is the weekly average of the effective federal funds rate during the week of the STBL survey and inflation is the quarterly growth rate in the core CPI (that is, the CPI net of food and energy prices). Our results are robust to using ex-post inflation as an alternative measure of expected inflation.

Our other main loan characteristic is based on loan size which we use as a proxy for firm size. Larger loans tend to correspond to larger borrowers, so our proxy allows us to discuss differences in the effect of monetary policy across firm sizes. To reflect thresholds in lending practices for firms of different sizes, we separate loan size into three categorical variables (jointly referred to as FIRM SIZE): "Small Firm" is a loan with a size of \$100,000 or less, "Medium Firm" is a loan with a size of more than \$100,000 and no greater than \$1 million, and "Large Firm" is a loan with a size greater than \$1 million.<sup>9</sup>

We collect additional loan-specific information from the STBL to control for other aspects of loan heterogeneity. We control for the maturity of the loan to ensure that the composition of short-term and long-term loans is not changing in a way that might be correlated with commitment status. The dummy variable "Short Maturity" (also referred to as SHORT MATURITY) indicates a loan with a maturity of less than one year. The maturity of loans in our sample ranges from 1 day to 10 years, with 83.5% having a maturity of less than one year. We also control for two other variants of loan maturity. Let "Demand" indicate a loan for which the bank has the right to demand repayment at any time or the borrower has the right to prepay without cost<sup>10</sup> and let "Overnight" indicate loans with a recorded maturity of one day. In addition to maturity, we control for several other loan contract terms which may be correlated with commitment status. Specifically, we include indicators for whether the loan is collateralized and whether the loan has a floating interest rate. These loan terms

<sup>&</sup>lt;sup>9</sup> Erel (2006) uses credit size (the maximum of loan size and commitment limit) as a proxy for firm size in the STBL data. This proxy would have a bias in our specification due to its positive correlation with a loan's commitment status (by construction).

<sup>&</sup>lt;sup>10</sup> We assign demand loans a maturity of one day.

may be endogenously determined when a loan or commitment is negotiated. Our results are robust to the omission of these variables.

The STBL identifies the bank originating each loan, which allows us to match the STBL data to the quarterly Call Report data for each lender. We focus on total bank assets as the primary bank characteristic in our empirical specification.<sup>11</sup> This provides an interesting question of whether bank size is related to the relative importance of the bank lending channel and the balance sheet channel. It is also possible that the prevalence of commitment loans could depend on the size of the bank making the loan, so we include the natural log of a bank's total assets as a control. Another advantage of the STBL match with bank identities is the opportunity it provides to use a bank-specific fixed-effects specification.

We also control for changes in macroeconomic factors other than monetary policy. Our secondary measure of credit conditions is the Baa-Aaa spread, which measures the risk premium in the corporate bond market. This variable may capture some of the changes in the probability of default at the economy-wide level. We also include the yield curve slope as a way of controlling for the term premium in the treasury market. For the real economy, we include GNP growth and quarterly unemployment. These measures are intended to control for potential changes in the business cycle that may be correlated with credit tightness. Lastly, a linear time trend controls for both unobserved changes in the macroeconomic environment and the upward trend in the number of commitment loans relative to spot loans.

#### 4. Empirical methodology: Identifying changes in loan supply

The effect of tight monetary policy on bank lending differs in the two theories of the credit channel. According to the theory of the bank lending channel, aggregate bank loan supply is reduced during periods of tight monetary policy. In the theory of the balance sheet channel, bank loan supply is reallocated from small firms to large firms.

We use two empirical specifications to identify changes in bank lending in response to changes in monetary policy. Our first empirical specification is based on the *proportion* of commitment loans to spot loans while the second identifies the driving factors in these proportional changes by focusing on the *quantities* of commitment loans and spot loans. Both

<sup>&</sup>lt;sup>11</sup> Future research could consider other bank characteristics, such as bank liquidity and capitalization.

are designed to identify changes in bank loan supply during periods of tight monetary policy and the quantity specification also allows us to estimate demand changes.

We use the loan-level contract features of the STBL to empirically differentiate the two theories of the credit channel. By using loan level data, we can control for some borrower heterogeneity through individual loan contract terms. Our main focus is on the difference between commitment loans and spot loans. The commitment status of each loan is the key component in our mechanism for identifying changes in loan supply. The basic idea is to use commitment lending as a baseline for measuring changes in the supply of spot loans. Our comparison of commitment lending and spot lending centers around a comparison by loan maturity and loan size. The ability to identify changes in loan supply helps us estimate the specific contributions of the bank lending channel and the balance sheet channel.

Loan commitments limit the ability of banks to control the amount of lending in a given period. If a firm has an existing line of credit, then it can draw down funds whenever it has a demand for credit.<sup>12</sup> This gives control over the amount of commitment lending to borrowers, meaning that changes in commitment lending reflect changes in loan demand. Although the writing of commitment contracts can vary the supply of commitment loans over longer periods of time, our data indicate that most drawdowns<sup>13</sup> occur months after the signing of a commitment. Starting in the third quarter of 2003, we have data on the date a commitment was signed if a loan is made under commitment. These data show that only 25% of commitment borrowings are from commitment agreements that were signed in the three months prior to the draw down, with 57% over 6 months old and 31% over one year old. Thus, the ability of banks to adjust the supply of commitment loans is likely to be limited in the short term, which means that changes in commitment lending can be used as a measure of changes in loan demand. To the extent that there is an impact on supply, it means that we may underestimate the effect of changes in supply and overestimate the effect of changes in demand.

Under the assumption that loan demand is relatively consistent across the two types of borrowing, changes in loan demand should be observed as joint movements in the levels of

<sup>&</sup>lt;sup>12</sup> As noted above, commitments typically include a material adverse change (MAC) clause which protects the bank in the event that a firm's condition deteriorates. However, these clauses are not frequently exercised.

<sup>&</sup>lt;sup>13</sup> A "drawdown" is a loan made to a firm with a loan commitment. The firm draws funds from its available credit.

spot loans and commitment loans. If the two types of lending increase together, this can be attributed to an increase in demand and, likewise, if the levels of the two types decrease together, this can be attributed to a decrease in demand. Such changes in response to monetary policy operate through the interest-rate channel rather than a credit channel. Although monetary policy tightening should slow loan demand, there is no reason that overall lending must decrease. A tightening reduces demand from what it otherwise would be, and since tightenings are more likely to occur when the economy is strong, loan demand may still increase when interest rates are rising.

Unlike with commitment loans, banks can immediately adjust the supply of spot loans. Banks adjust the supply of spot loans by simply raising or lowering their selection requirements for loan applications. The difference in the adjustability of loan supply for commitment loans and spot loans provides a mechanism for identifying changes in spot loan supply. When the quantity of spot loans changes *relative* to the quantity of commitment loans, we interpret this as a change in the supply of (spot) loans. In other words, we use the quantity of commitment loans as a baseline against which to measure changes in the supply of loans. If the quantity of spot loans decreases relative to the quantity of commitment loans, we interpret this as a reduction in the supply of loans. On the other hand, if the quantity of spot loans increases relative to the quantity of spot loans increases relative to the quantity of spot loans. We interpret this as an increase in the supply of loans.

#### Logit Analysis

Our first empirical specification measures changes in the propensity for a loan to be under commitment. This approach is the loan-level equivalent of analyzing changes in the proportion of loans that are taken under commitment. This fits our loan supply identification strategy, which analyzes changes in spot loans relative to commitment loans.

The baseline model for testing our hypotheses is the fixed-effects logit model

COMMIT = f (FUNDS RATE, SHORT MATURITY, FIRM SIZE, control variables),(2)

where COMMIT is the dummy variable indicating whether the loan is a drawdown under a loan commitment and where we control for bank fixed effects.

We estimate equation (2) to provide an estimate of how the credit channel operates at the aggregate level based on the number of loan originations. However, our loan-level data allows us to analyze bank lending beyond an aggregate measure of the number of loans. The bank lending channel and balance sheet channel may have a reallocative effect that might be hidden by this type of analysis.

An operative bank lending channel implies a reduction in the stock of loans during periods of tightening monetary policy, which may occur is the flow of lending slows or if a bank reduces the average maturity of the loans it issues. When a bank reduces the maturity of its new loans, it gradually reduces the size of its loan portfolio over time. The rate of "outflow," or the rate at which loans are paid off, increases, which reduces the number of loans in the portfolio at any given time. Dividing the sample by loan maturity allows us to examine how the maturity structure of the new loans measured by the STBL changes with monetary policy adjustments. We split the sample into two groups: short-maturity loans (SHORT MATURITY = 1, i.e., less than one year maturity) and long-maturity loans (all others). We choose the one year split, because this typically corresponds to the accounting distinction of short-term debt and long-term debt.

The effect of the balance sheet channel is that banks reallocate loan supply from small firms to large firms during periods of tight monetary policy. Small firms tend be riskier than large firms due to less diversification and, often, a riskier balance sheet, so the balance sheet channel predicts that a reduction in net worth due to a monetary contraction causes a shift in bank lending from small firms to large firms. To test this, we divide the sample by the loan size, our proxy for firm size. We use three size classes – small firms, medium firms, and large firms – to capture different aspects of the lending choice. Loans for small firms are often based on different criteria than those for larger firms (Berger and Udell (2006)) while the largest firms have more non-bank borrowing options than smaller firms. Our results are robust to different loan size splits.

Lastly, we analyze the full 6-way split by maturity and firm size. We analyze both dimensions together in the final set of regressions, because the response of lending to

monetary policy along both dimensions can be used to further differentiate the effects of the two channels.

The control variables include the Baa-Aaa spread, the yield curve slope, the log of bank assets, GNP growth, unemployment, and a time trend. We also include loan level control variables which indicate whether a loan is collateralized, floating rate, a demand loan, or a overnight loans. As noted, the regression includes fixed effects; therefore, the other variables explain changes of the probability of commitment for loans within a given bank. All the reported p-values in the regressions are robust and are corrected for bank-level clustering.

Our measure of monetary policy is the real federal funds rate (FUNDS RATE), with a higher rate indicating a policy tightening (Bernanke and Blinder (1992)). In our first analysis, we analyze whether the probability of a loan being under commitment increases or decreases with the funds rate. A positive coefficient on the funds rate indicates a reduction in loan supply when monetary policy while a negative coefficient implies the opposite. According to the theory of the bank lending channel, banks reduce their loan supply during periods of tight monetary policy, so a positive coefficient is consistent with the theory. The balance sheet channel does not have a clear prediction for the aggregate level of bank loan supply, because it is a theory about the reallocation of credit.

When we split the sample by loan maturity, a reallocation from long-term loans to short-term loans would be support for the bank lending channel. We would see this if the coefficients on the funds rate were more positive (or less negative) for long-term loans than for short-term loans. Again, the coefficients on the funds rate in these regressions do not allow us to test the balance sheet channel.

We use the firm size sample splits to analyze the balance sheet channel. The balance sheet channel predicts that banks reallocate loan supply away from small firms and toward large firms. If the coefficient on the the funds rate is decreasing in firm size, then banks are increasing spot lending relative to commitment lending more for large firms than for small firms, consistent with the predictions of an operating balance sheet channel.

As stated above, our approach can be thought of as similar to the KSW approach. KSW analyze changes in bank debt relative to commercial paper (CP) debt. Like us, they assume that credit demand is relatively consistent across two different forms of borrowing. For their study, if the quantity of bank debt falls relative to CP debt following a contraction, then they interpret this as a reduction in bank loan supply. We apply a similar approach using loan commitments. If the quantity of spot loans falls relative to the quantity of commitment loans when monetary policy is tight, then we interpret this as a reduction in bank loan supply.

There are several advantages to examining changes in spot lending relative to commitment lending as an approach for identifying changes in loan supply. First, it can be applied both to small firms and large firms. The analysis in KSW is implicitly restricted to the largest borrowers, since only these firms have access to the commercial paper market while the alternative in Oliner and Rudebusch (1996) has similar issues, since they must assume that a catch-all category called "other debt" really measures bank loan substitutes for small firms. Second, our approach does not rely on other measures of real activity for firms, such as inventories (as in Kashyap, Lamont, and Stein (1994)). The commitment status approach compares the relative amounts of two forms of debt, where we can more closely separate changes in the demand and supply of debt.

#### Quantity Analysis

In our second empirical specification, we analyze the total quantities of commitment loans and spot loans. This provides us with information about the underlying elements behind the logit specification. We analyze the quantities of each type of loan to examine whether the ratio of the two loan types changes primarily due to changes in the numerator or changes in the denominator. Specifically, we identify the levels of commitment and spot loans separately and discuss their respective changes independently. These findings provide additional insight into how the two channels operate.

To look at quantity levels, we must depart from the loan-level observations of the logit specification. In the quantities specification, we use bank-quarter observations. We aggregate the loans by summing the number of loans made by a bank in a quarter. The dependent variable is the number of loans made by a bank in a quarter, normalized by the mean number of loans for the bank in the sample as a whole. This normalization implies that variation in the dependent variable is a bank-level percentage variation in the number of loans relative to a bank-level mean. Our baseline model for bank-level loan quantities is:

Loan Quantity = 
$$\alpha_0 + \alpha_1$$
 FUNDS RATE +  $\beta$  \* control variables +  $\epsilon$ . (3)

where Loan Quantity represents either the normalized level of spot loans or commitment loans. The control variables include all of the same regressors as the logit specification except for the loan-level variables. We also include bank fixed effects to control for differences in the level of lending across banks. As with the logit results, all the reported p-values in the regressions are robust and are corrected for bank-level clustering.

We analyze the change in lending quantities for the sample as a whole and also for subsamples split by loan maturity and firm size. Again, these two dimensions allow us to differentiate the unique contributions of the bank lending channel and the balance sheet channel.

Again, the key independent variable is the real federal funds rate variable. A positive coefficient on the funds rate indicates that the number of loans increases with the tightness of monetary policy. Likewise, a negative coefficient indicates that the number of loans decreases with tight monetary policy. We use changes in the quantity of commitment loans as our measure of changes in loan demand. As stated above, this is based on the assumption that the supply of commitment loans adjusts slowly to changes in to the funds rate. We interpret changes in spot loans relative to commitment loans as a change in loan supply.

#### 5. Results

#### Full Sample

Table 2 reports the results for our logit specification based on the full sample of all commitment and spot loans. Our first set of results analyzes the total flow in the number of loans as a first pass at identifying a change in loan supply. This approach treats all loans equally and identifies the relative changes in the total number of spot loans and the total number of commitment loans. Under this simplifying assumption, we can analyze the aggregate flow of new loans as a measure of aggregate loan supply. The bank lending channel prediction is that aggregate bank loan supply decreases when monetary policy is tight. Our results show that the coefficient on the funds rate is -0.025, which is negative but insignificant. This indicates that the number of spot loans does not significantly increase or decrease relative to the number of commitment loans when the funds rate increases. Based on

this result, there does not appear to be a reduction in bank loan supply when the funds rate increases.

The quantity results for the aggregate number of loans in our second empirical specification are shown in Table 3. Column 1 shows the results for all loans (including commitment and spot). The coefficient on the funds rate is 0.014, which is significant at the 1% level, indicating that the number of loans increases with the tightness of monetary policy. This relationship could be due to an increase in either loan demand or loan supply. Columns 2 and 3 show the results for the split samples of commitment loans and spot loans, respectively. These quantity results indicate that when the federal funds rate increases, there is no significant change in commitment loans is relatively sticky, this finding for commitment lending indicates that loan demand does not change significantly with tight monetary policy. The coefficient on the funds rate in the spot loan regression is 0.027, which is significantly larger than the coefficient in the commitment regression. This indicates that the percentage increase in the amount of spot lending is significantly greater than the percentage increase in the amount of commitment lending.<sup>14</sup>

The aggregate logit and quantity results together indicate a slight increase in loan supply with increased interest rates. Although this supply shift is not significant in the logit regressions, the positive sign on the funds rate is more indicative of an increase in loan supply than a decrease in loan supply. Additionally, the quantity results show a significant increase in spot lending relative to commitment lending. Therefore, our initial results do not support the bank lending channel prediction that loan supply decreases in periods of tight monetary policy. However, these results are based on the assumption that all loan originations have equal weight in a bank's loan portfolio.

#### By Loan Maturity

In our first sample split, we analyze changes in bank lending by loan maturity to test a potential mode of operation for the bank lending channel. Banks can control the size of their loan portfolio over time by varying the maturity of their loan originations. If a bank reduces

<sup>&</sup>lt;sup>14</sup> The sample sizes are different, because not all banks make both commitment loans and spot loans in every quarter.

the maturity of its loan originations, this reduces the length of time that its capital is invested in each project. Although banks may not significantly cut back on the number of loans originations in response to tight monetary policy, a reduction in the maturity of loan originations reduces the outstanding stock of loans over time. This is a reduction in bank loan supply, as predicted by the bank lending channel.

The results in Table 4 show the coefficients for the logit regression split by short and long loan maturity. Column 1 shows the results for the sample of loans with a maturity less than a year. The coefficient on the real federal funds rate is -0.035 and is significant. Because the proportion of spot loans increases relative to the proportion of commitment loans, this indicates that the supply of short-term loans increases with the funds rate. Column 2 shows the coefficients for the regression on loans with at least a year of maturity. In this case, the coefficient on the funds rate is positive, but insignificant. Although the coefficient itself is not significant in this regression, it is significantly larger than the funds rate coefficient in the short-term regression.

The results for the quantities specification split by short-term and long-term are shown in Table 5. For short-term commitment loans, the coefficient on the funds rate is 0.001 and is insignificant, and for short-term spot loans, the coefficient is 0.030 and significant. Among short-term loans, it appears that commitment lending does not respond to monetary policy whereas spot lending increases during periods of tight monetary policy. For long-term commitment loans, the coefficient on the funds rate is 0.055 while for long-term spot loans, the coefficient is 0.025. Both are statistically significant and the difference between the coefficients of 0.030 is also significant. The increase in the quantity of commitment loans indicates an increase in demand for long-term loans.<sup>15</sup> In this case, it is the spot loans that minimally respond to monetary policy and the commitment loans that significantly increase when monetary policy is tight.

The logit results and the quantity results give a consistent picture. For short-term loans, the spot lending goes up with the federal funds rate while commitment lending does not change, which is consistent with a decrease in the proportion of commitment loans we find. Together, the short-term results show that spot lending increases relative to commitment

<sup>&</sup>lt;sup>15</sup> As Gertler and Gilchrist (1994) have shown, this increased demand for debt may be due to an increased need to carry inventories. When firms face reduced sales due to reduced consumer demand, they may increase borrowing in order to maintain production and inventories.

lending, which we expect to lead to an increase in the supply of short-term loans. For longterm loans, the number of commitment loans increase more than the number of spot loans when the funds rate increases. This is consistent with the positive (albeit insignificant) coefficient on the funds rate in the long-term loan logit regression, Together, these results indicate that long-term commitment lending might weakly increase relative to long-term spot lending, which we interpret to be a slight reduction in the supply of long-term loans. The combined results indicate a reallocation of loan supply from long-term lending to short-term lending during periods of tight monetary policy.

These results for loan maturity are generally consistent with our hypothesis based on the bank lending channel. According to the bank lending channel, banks reduce their aggregate loan supply during periods of tight monetary policy. We did not find evidence for this reduction in our full-sample regressions; however, when we split the sample by maturity, we find that banks tend to reallocate their loan supply from long-term loans to short-term loans. This indicates a move toward greater liquidity in their portfolio which could be due to increased costs of bank financing. Over time, a reduction in the maturity of originations reduces the overall size of a bank's loan portfolio. This suggests that the bank lending channel may operate through a reduction in the maturity of loan originations.

#### By Firm Size

We analyze loan supply by firm size, because the balance sheet channel predicts a reallocation of loan supply across firms of different sizes. The advantage of the loan commitment approach to identifying loan supply is that we can analyze changes in loan supply to small firms and large firms using the same method for each. This can help us identify the contribution of the balance sheet channel within bank lending. In our framework, the theory of the balance sheet channel translates into the prediction that banks reallocate loan supply from small firms to large firms when the funds rate is high.

The logit results in Columns 1 through 3 of Table 6 present separate regressions of (2) for each of the three size categories. The results for small and medium firms do not indicate much movement in the proportion of commitment loans with the funds rate. The coefficients on the funds rate for small firms and medium firms are both statistically insignificant and economically small. These results indicate that the probability of a small or medium loan

being under commitment is not affected by movements in the real federal funds rate changes. Column 3 reports the results for a regression of (2) for large firms. The coefficient on the funds rate for large firms is a significant -0.054, indicating that, for large firms, the proportion of spot loans increases relative to commitment loans when the funds rate increases. Based on our identification of loan supply, these results indicate a difference in the response of loan supply across firms of different sizes. When monetary policy is tight, banks appear to increase the loan supply to large firms, but not to small firms or medium firms. An F-test for all three firm size interactions rejects the possibility that the coefficients are statistically identical. The results clearly indicate that the relationship between commitment status and the funds rate differes across borrower size categories.

The quantity results by firm size, as shown in Table 7, generally coincide with the logit results. For commitment lending, the coefficients on the funds rate are not significant, although there is a slight upward trend in firm size. However, spot lending increases significantly with the funds rate for small firms and the increase in spot lending to large firms is even greater. For spot lending to small firms, the coefficient on the funds rate is 0.033 while for large firms, it is an even larger 0.093. These two values are significantly different. These quantity results indicate that the increased proportion in spot lending to large firms is driven by a significant increase in spot lending with the funds rate.

The theory of the balance sheet channel predicts that banks reallocate credit away from small firms and toward large firms during periods of tight monetary policy. Our findings in the logit specification and the quantities specification are both consistent with this prediction. Although we do not find evidence for a reduction in the supply of loans to small firms, we do find evidence for an increase in the supply of loans to large firms relative to small firms. These results by firm size provide an initial indication of how the balance sheet channel operates.

#### By Loan Maturity and Firm Size

Table 8 shows differences in the effect of monetary policy by both loan maturity and firm size. This analysis gives a 6-way representation of the role of firm size across short-term and long-term loans. Splitting the loans two ways for maturity (short and long) and three ways for firm size (small, medium, and large) identifies six categories of loans. This analysis

combines our previous analyses by maturity and firm size so that we can further specify the unique contributions of the bank lending channel and the balance sheet channel.

The logit results from regressions of (2) on the 6-way split of SHORT MATURITY and FIRM SIZE are shown in the six columns of Table 8. Using this specification, we can identify the derivative with respect to the funds rate for each of these six categories. This specification also controls for differences in the control variable coefficients across the six regressions.

Among the short-term loans, the coefficients on the funds rate are negative for all firm sizes. The coefficients range from -0.019 for small firms to -0.074 for large firms, all significantly different from zero. This shows that the increase in short-term spot loans relative to short-term commitment loans is greatest for large firms. The combined results indicate that large firms experience an increased allocation in short-term loan supply relative to small firms.

For long-term loans, none of the coefficients on the federal funds rate are significant and there is no significant pattern across the firm sizes. However, it is important to note that the coefficients on the funds rate in the long-term loan regressions are each significantly more positive than the coefficients on the funds rate in the short-term loan regressions of the same firm size. For small firms, the long-term funds rate coefficient is 0.037, which is 0.056 greater than the short-term coefficient, a difference that is significant. This implies a shift from long-term to short-term lending for small firms. The shift is even greater for medium and large firms.

In the quantity regressions shown in Table 9, the differences between the short-term loans and the long-term loans are again clear. For the short-term loans, the coefficient on the funds rate is positive in each of the spot loan regressions. In the small firm spot loan regression, the coefficient is 0.031, while for the large firm spot loan regression, it is 0.125, with both significantly positive. This difference of 0.094 between the coefficients is also significant. This shows that the increase in short-term spot lending is greatest for large firms, which is consistent with the logit results. In the regressions for short-term commitment loans, none of the coefficients on the funds rate are statistically different from zero. For the long-term loans, the coefficients on the funds rate in the commitment loan regressions are all positive and significant, although they are not statistically different from each other.

According to our identification, we interpret this as an increase in loan demand. The coefficients on the funds rate for long-term spot lending are all insignificant.

The quantity regressions add a further layer of understanding to the 6-way maturity and size split. It is clear from these regressions why the proportion of short-term spot lending increases with the funds rate more for large firms. This occurs primarily because of the sizable increase in spot lending to large firms. Short-term commitment lending is relatively flat across the different size groups. For the long-term loans, the logit specification shows that the proportion of long-term commitment lending may increase with the funds rate for all firms and the quantity regressions also show a statistically significant increase in commitment lending relative to spot lending. These results show that there is limited variation in longterm lending across firms of different sizes.

Our 6-way split of the sample by maturity and firm size provides additional insight into the operation of the two channels. In this step, we show that the increase in loan supply to large firms is primarily among short-term lending. Therefore, the complete picture seems to indicate a reduction in loan maturity, which is consistent with the bank lending channel, and a shift in short-term lending toward large firms, which is consistent with the balance sheet channel.

#### By Bank Size

The credit channel literature has focused on bank size as an important factor in determining the sensitivity of loan supply to monetary policy. Although Romer and Romer (1990) argue that the lending channel has been weakened due to the availability of nondeposit sources of funds for banks, Stein (1998) shows that uninsured forms of bank finance are subject to adverse selection. Therefore, smaller banks may have significant costs of raising external funds during periods of tight monetary policy. Kashyap and Stein (1995) analyze the sensitivity of bank lending to monetary policy for different bank sizes under the hypothesis that large banks face lower costs of external financing. The authors find that the lending of large banks is not affected by monetary policy while the lending of small banks significantly declines when the nominal federal funds rate increases.

To test whether, because of differential funding costs or other reasons, a bank's size affects the way it reacts to monetary policy changes, we split our sample between large and small banks. "Small banks" are banks with total assets less than or equal to \$10 billion and "large banks" are banks with total assets greater than \$10 billion. This threshold allows us to insure that the large banks in our sample have the low-cost access to financing present in the arguments of Stein (1998) and Kashyap and Stein (1995). In our full sample, there are 183 small banks and 112 large banks.

Table 10 presents results for the logit and quantity regressions when the sample is split by bank size. The results in the table also include the division of the sample by maturity and firm size. To simplify the presentation, we give the coefficient for the federal funds rate variable only, but we include all the controls used in the prior regressions.

Our results *do not* indicate that commercial lending is more responsive to monetary policy at small banks than at large banks. We find evidence consistent with an increase in the federal funds rate reducing the supply of long-maturity loans at banks of both sizes, as the coefficients in the logit regressions and in the commitment quantity regressions are positive and often significant. There are no significant differences between small and large banks in the impact of monetary policy changes on long-maturity loans. To the extent that bank size matters, we find a stronger reaction in short-maturity loans by large banks. When policy tightens, large banks increase the supply of short-maturity loans while there is no significant reaction at small banks. Overall, the bank lending channel seems to operate at both groups of banks and, if anything, more strongly at large banks.

The splits by firm size in Table 10 show some evidence of the balance sheet channel operating through both small and large banks. Although the statistical power is limited due to small sample sizes, the funds rate coefficients in the logit regression increase monotonically with firm size for long-maturity loans at small banks. This weakly indicates that the reduction in small banks' long-term loan supply may be greater for large firms than small firms. Large banks, on the other hand, appear to significantly shift their short-term loan supply from small firms to large firms. This finding is consistent with the theory that large banks are not as effective as small banks at overcoming information asymmetries in lending (Stein (2002)). In the theory of the balance sheet channel, lenders shift their lending from small firms to large firms during periods of tight monetary policy due to the increased agency costs associated with firms of low net worth. If large banks are less capable of mitigating agency costs than

small banks, this implies that an increase in agency costs for small firms would cause a greater shift in large banks' loan supply from small firms to large firms.

Our findings show that the credit channel works through both small and large banks, with some evidence that the strength of the channel is greatest for large banks. This is a surprising result, given previous findings (e.g. Kashyap and Stein (1995, 2000)) which indicate that large banks are less sensitive to the credit channel. This difference in results may arise because Kashyap and Stein do not examine loan maturity and firm size. Our analysis shows that the full sample results can hide significant shifts in loan maturity and borrower type over the monetary policy cycle. We find that the effect of the credit channel for large banks is primarily in the reduction of loan maturity and the reallocation from small firms to large firms.

#### Control Variables

Lastly, the coefficients on the control variables in the logit regressions provide some information about the relative probability of commitment lending and spot lending. The results for the control variables are relatively consistent across the various maturity and size specifications. When a loan is collateralized, has a floating rate, is repayable/recallable on demand, or is made overnight, the probability of the loan being made under commitment tends to be higher. The coefficients on bank size, on the other hand, tend to be negative, suggesting that large banks are less likely to make commitment loans, though the coefficients are generally insignificant. The macro variables suggest that commitment status is not systematically related to other financial or real-economy factors. The Baa-Aaa spread, the yield curve slope, GNP growth, and unemployment are generally insignificant. Finally, the coefficient on the time trend is positive and significant, indicating a trend toward more loans being taken under commitment over time.

#### 6. Conclusion

The existence of a credit channel operating beyond the traditional interest-rate channel of monetary policy has been well documented (see Mishkin (2007) for a review). Less settled is whether the credit channel operates by variations in aggregate lending (the bank lending channel) or through changes in the aggregate risk of borrowers (the balance sheet channel).

Previous studies have found support for a single channel (e.g. Kashyap and Stein (2000) for the bank lending channel and Bernanke, Gertler, and Gilchrist (1996) for the balance sheet channel) or claimed that they could not differentiate the two channels (e.g. Morgan (1998)). The main contribution of our paper is to identify the unique operation of the bank lending channel and the balance sheet channel in the transmission of monetary policy.

We use an innovative approach for identifying changes in loan supply. The identification of loan supply using cross-sectional data is one of the central issues in analyzing the credit channel. Often, studies rely on external debt to benchmark changes in supply, but are subject to criticism based on the choice of external debt and because it is difficult to determine the external debt options of small firms (see the debate between Kashyap, Stein, and Wilcox (1993, 1996) and Oliner and Rudebusch (1996)). We use an alternative approach which is entirely internal to bank lending, taking advantage of the fact that banks make both commitment and spot loans to small and large firms.

Commitment loans and spot loans differ in how much flexibility they leave banks to quickly adjust supply when conditions change. It is difficult for a bank to change the supply of loan commitments. Other than the case of a firm undergoing a material adverse change, a bank cannot readily deny credit to a firm which has an existing commitment. Additionally, there is evidence that the majority of commitment loans are made under commitments signed at least 6 months prior to the drawdown. Therefore, we treat the commitment loan supply curve as fixed with respect to the interest rate cycle. The supply of spot loans, on the other hand, can be adjusted instantaneously. A bank can decide to tighten its credit standards and thereby deny an applicant who would have gotten a loan previously.

Since we pin down the commitment loan supply curve, changes in commitment lending when monetary policy changes reflect changes in loan demand. If demand is relatively consistent across commitment lending and spot lending, this allows us to determine changes in the supply of spot loans. We do this by examining how spot lending moves *relative* to commitment lending. A decrease in the number of spot loans relative to the number of commitment loans indicates a reduction in the supply of spot loans.

We test the operation of the credit channel using two different specifications. Our first specification examines the probability that a loan is taken under commitment versus as a spot loan. This specification, which is based on loan level data, allows us to control for some firm

heterogeneity using loan contract terms. The second specification analyzes the quantity of commitment lending and spot lending at the bank level. This allows us to decouple the commitment-to-spot ratio to analyze the separate changes in the numerator and denominator.

We focus our analysis on changes in lending by loan maturity and firm size, because these dimensions allow us to differentiate the operation of the bank lending channel and the balance sheet channel. The bank lending channel can operate through changes in loan maturity since if a bank increases the outflow of loans by requiring a short-term repayment and the bank does not increase the inflow of loan originations, this will gradually reduce the bank's stock of loans over time. This is equivalent to a reduction in the supply of bank loans, as predicted by the bank lending channel. The balance sheet channel operates through a reallocation of loan supply from small firms to large firms. We analyze changes in bank lending by firm size to test for the presence of a balance sheet channel in bank lending.

Our results provide new insight into the operation of the credit channel. We do not find evidence for a decrease in the aggregate flow of spot loan originations relative to commitment loan originations when the funds rate is high; however, these results are probably not the best indicator of loan supply. Both of the channels internal to the credit channel can operate through reallocations of loan supply. Our analysis by loan maturity reveals that banks reduce the maturity of their loan originations during periods of tight monetary policy. This indicates that banks increase the frequency of loan repayment, which reduces the size of banks' loan portfolios over time. This can effectively reduce loan supply, as predicted by the bank lending channel. Our analysis by firm size shows that banks increase their lending to large firms during periods of tight monetary policy. An analysis by maturity and firm size shows that this effect is driven by a reallocation of short-term lending. Consistent with the balance sheet channel, banks reallocate short-term lending from small firms to large firms.

The full set of our results helps us identify the unique contribution of two different theories of the monetary policy transmission mechanism. Although previous papers in the literature have identified the credit channel, or have even found evidence for the bank lending channel or balance sheet channel separately, the effects of the two channels have not been clearly evaluated. This was partly due to the difficulty of identifying changes in loan supply for both small firms and large firms. Our paper uses a unique identification strategy to differentiate the operation of the two channels within the credit channel. By separating commitment lending from spot lending, we can identify changes in loan supply by loan maturity and firm size. Our results show that banks reallocate total loan supply from long-term to short-term and short-term loan supply from small firms to large firms. These findings suggest that, within the credit channel of monetary policy, the bank lending channel causes banks to reduce the maturity of their loans and the balance sheet channel causes banks to reallocate their short-term lending toward large firms.

#### References

- Ashcraft, A. 2006. New evidence on the lending channel. Journal of Money, Credit and Banking 38, 751-775.
- Avery, R., Berger, A., 1991. Loan commitments and bank risk exposure. Journal of Banking and Finance 15, 173-92.
- Berger, A., Udell, G., 1992. Some evidence on the empirical significance of credit rationing. Journal of Political Economy 100, 1047-77.
- Berger, A., Udell, G., 2006. A more complete conceptual framework for SME finance. Journal of Banking and Finance 30, 2945-2966.
- Bernanke, B., Blinder, A., 1988. Credit, money, and aggregate demand. American Economic Review 78, 435-439.
- Bernanke, B., Blinder, A., 1992. The federal funds rate and the channels of monetary transmission. American Economic Review 82, 901-921.
- Bernanke, B., Gertler, M., 1989. Agency costs, net worth, and business fluctuations. American Economic Review 79, 14-31.
- Bernanke, B., Gertler, M., 1995. Inside the black box: The credit channel of monetary policy transmission. Journal of Economic Perspectives 9, 27-48.
- Bernanke, B., Gertler, M., Gilchrist, S., 1996. The financial accelerator and the flight to quality. The Review of Economics and Statistics 78, 1-15.
- Erel, I., 2006. The effect of bank mergers on loan prices: Evidence from the U.S. Working paper, Ohio State University.
- Gertler, M., Gilchrist, S., 1994. Monetary policy, business cycles, and the behavior of small manufacturing firms. Quarterly Journal of Economics 109, 309-40.
- Jayaratne, J., Morgan, M., 2000. Capital market frictions and deposit constraints at banks. Journal of Money, Credit and Banking 32, 74-92.
- Kashyap, A., Lamont, O., Stein, J., 1994. Credit conditions and the cyclical behavior of inventories. Quarterly Journal of Economics 109, 565-592.
- Kashyap, A., Stein, J., 1995. The impact of monetary policy on bank balance sheets. Carnegie-Rochester Conference Series on Public Policy 42, 151-195.

- Kashyap, A., Stein, J., 2000. What do a million observations on banks say about the transmission of monetary policy? American Economic Review 90, 407-428.
- Kashyap, A., Stein, J., Wilcox, D., 1993. Monetary policy and credit conditions: Evidence from the composition of external finance. American Economic Review 83, 78-98.
- Kashyap, A., Stein, J., Wilcox, D., 1996. Monetary policy and credit conditions: Evidence from the composition of external finance: Reply. American Economic Review 86, 310-314.
- King, S., 1986. Monetary transmission: Through bank loans or bank liabilities? Journal of Money, Credit and Banking 18, 290-303.
- Kishan, R., Opiela, T., 2000. Bank size, bank capital, and the bank lending channel. Journal of Money, Credit and Banking 32, 121-141.
- Lang, W., Nakamura, L., 1995. "Flight to quality" in bank lending and economic activity. Journal of Monetary Economics 36, 145-164.
- Mishkin, F., 2007. The economics of money, banking, and financial markets. Boston : Pearson Addison Wesley, c2007. 8th ed.
- Morgan, D., 1998. The credit effects of monetary policy: Evidence using loan commitments. Journal of Money, Credit and Banking 30, 102-18.
- Oliner, S., Rudebusch, G., 1996. Monetary policy and credit conditions: Evidence from the composition of external finance: Comment. American Economic Review 86, 300-309.
- Romer, C., Romer, D., 1990. New evidence on the monetary transmission mechanism. Brookings Papers on Economic Activity, 149-198.
- Sofianos, G., Wachtel, P., Melnik, A., 1990. Loan commitments and monetary policy. Journal of Banking and Finance 14, 677-89.
- Stein, J., 1998. An adverse selection model of bank asset and liability management with implications for the transmission of monetary policy. Rand Journal of Economics 29, 466-486.
- Stein, J., 2002. Information production and capital allocation: Decentralized versus hierarchical firms." Journal of Finance 57, 1891-1921.
- Stiglitz, J., Weiss, A., 1981. Credit rationing in markets with imperfect information. American Economic Review 71, 393-410.

#### **TABLE 1: Descriptive Statistics**

The mean and standard deviation (in brackets) for the variables in our loan-level regressions The loan data are from the Survey of Terms of Business Lending (STBL). The macroeconomic data are drawn from Fred II at the St. Louis Federal Reserve. (The macroeconomic statistics are only shown for the full sample.) The sample covers 1982:Q3 - 2006:Q1.

		Total	Short Maturity	Long Maturity	Small Firm	Medium Firm	Large Firm
Commit	A dummy variable: a loan made under commitment	0.816	0.819	0.796	0.801	0.857	0.791
	,	[0.387]	[0.385]	[0.403]	[0.399]	[0.351]	[0.407]
Real Federal Funds Rate	Federal funds rate - current inflation (not seasonally adjusted)	2.142					
		[1.977]					
Short Maturity	A dummy variable: maturity < 1 year	0.835			0.864	0.788	0.785
-		[0.371]			[0.343]	[0.409]	[0.411]
Long Maturity	A dummy variable: maturity ≥ 1 year	0.165			0.136	0.212	0.215
		[0.371]			[0.343]	[0.409]	[0.411]
Small Firm	A dummy variable: loan size ≤ \$100K	0.626	0.647	0.516			
		[0.484]	[0.478]	[0.500]			
Medium Firm	A dummy variable: loan size > \$100K and ≤ \$1M	0.284	0.267	0.366			
		[0.451]	[0.443]	[0.482]			
Large Firm	A dummy variable: loan size > \$1M	0.090	0.086	0.119			
		[0.286]	[0.280]	[0.323]			
Baa-Aaa Spread	The Baa yield - Aaa yield	0.949					
		[0.333]					
Yield Curve Slope	The 10-year treasury yield - the 3-month treas. yield	1.482					
		[1.242]					
Collateral	A dummy variable: a secured loan	0.739	0.731	0.765	0.779	0.743	0.440
		[0.439]	[0.443]	[0.424]	[0.415]	[0.437]	[0.496]
Floating	A dummy variable: a floating rate loan	0.865	0.871	0.835	0.887	0.900	0.606
		[0.341]	[0.335]	[0.371]	[0.316]	[0.300]	[0.489]
Demand	A dummy variable: a loan that can be repaid/recalled on demand	0.368			0.390	0.372	0.220
		[0.482]			[0.489]	[0.483]	[0.414]
Overnight	A dummy variable: a loan with a one day maturity	0.014			0.002	0.007	0.117
		[0.117]			[0.039]	[0.085]	[0.321]
Ln(Bank Assets)	Log of total bank assets	16.878					
		[1.490]					
GNP Growth	Quarterly growth rate in gross national product	3.206					
		[2.306]					
Unemployment	Percentage of unemployment	5.747					
		[1.279]					
Trend	A linear time trend	53.890					
		[26.373]					
Observations		258810	538780	427463	404938	366375	235394

The current "total" sample is a tenth of the full number of observations.

The short-maturity sample is a fourth of the short-maturity observations. The long-maturity sample is the full long-maturity sample. The small firm sample is a fourth of the small firm observations. The medium firm sample is a half of the medium firm observations. The large firm sample is the full large firm sample.

#### TABLE 2: Logit Results - Aggregate

A logit regression on a dummy variable indicating whether a loan is made under commitment. The regression also includes bank fixed-effects. The omitted category is long-maturity loans to small firms. Errors are robust and clustered by bank. Standard errors are shown in brackets with \*, \*\*, and \*\*\* indicating significance at 10%, 5%, and 1%, respectively.

	Aggregate
Real Federal Funds Rate	-0.025 [0.019]
Short Maturity	0.403 [0.124]***
Medium Firm	0.469 [.090]***
Large Firm	0.665 [.269]**
Baa-Aaa Spread	-0.016 [0.267]
Yield Curve Slope	0.0432 [0.045]
Collateral	0.44 [0.098]***
Floating	1.22 [0.130]***
Demand	0.369 [0.166]**
Overnight	0.595 [0.363]
Ln(Bank Assets)	-0.263 [0.250]
GNP Growth	0.006 [0.011]
Unemployment	-0.033 [0.050]
Trend	0.041 [0.009]***
Observations Pseudo R-Squared	258810 0.34

(The current sample is a tenth of the full number of observations. Other random samples have been checked for robustness.)

#### TABLE 3: Quantity Results - Aggregate

An OLS regression with bank fixed effects. The dependent variable in each column is the number of loans made by a bank in a given quarter divided by the mean number of loans made by the bank in the entire sample. In the "Commit" column, the dependent variable is based on the number of commitment loans made and, in the "Spot" column, the dependent variable is based on the number of spot loans made. Errors are robust and clustered by bank. Standard errors are shown in brackets with \*, \*\*, and \*\*\* indicating significance at 10%, 5%, and 1%, respectively.

	(1)	(2)	(3)	
	Total	Commit	Spot	
Real Federal Funds Rate	0.014	0.009	0.027	
	[0.005]***	[0.006]	[0.009]***	
Baa-Aaa Spread	0.082	0.057	0.26	
	[0.057]	[0.070]	[0.091]***	
Yield Curve Slope	-0.019	-0.026	-0.013	
	[0.007]**	[0.011]**	[0.013]	
Ln(Bank Assets)	0.847	0.773	0.63	
	[0.083]***	[0.094]***	[0.120]***	
GNP Growth	0.003	0.004	0.007	
	[0.004]	[0.005]	[0.005]	
Unemployment	-0.023	-0.029	-0.003	
	[0.013]*	[0.016]*	[0.025]	
Trend	-0.004	0.004	-0.022	
	[0.002]*	[0.003]	[0.003]***	
Observations	12043	11563	10047	
Adjusted R-Squared	0.30	0.30	0.12	

## TABLE 4: Logit Results By Maturity

A logit regression on a dummy variable indicating whether a loan is made under commitment. The regressions also include bank fixed effects. The omitted category is small firms. "Short Maturity" is the sample of loans with maturity less than a year and "Long Maturity" is the sample of loans with maturity of at least a year. Errors are robust and clustered by bank. Standard errors are shown in brackets with \*, \*\*, and \*\*\* indicating significance at 10%, 5%, and 1%, respectively.

	(1)	(2)
	Short	Long
	Maturity	Maturity
Real Federal Funds Rate	-0.035	0.035
	[0.020]*	[0.023]
Medium Firm	0.451	0.433
	[0.095]***	[0.089]***
Large Firm	0.71	0.512
	[0.296]**	[0.252]**
Baa-Aaa Spread	-0.087	0.154
	[0.317]	[0.238]
Yield Curve Slope	0.059	-0.016
	[0.047]	[0.047]
Collateral	0.52	0.161
	[0.102]***	[0.175]
Floating	1.077	1.606
	[0.159]***	[0.137]***
Demand	0.429	
	[0.177]**	
Overnight	0.634	
	[0.321]**	
Ln(Bank Assets)	-0.265	-0.306
	[0.294]	[0.177]*
GNP Growth	0.007	0.018
	[0.013]	[0.010]*
Unemployment	-0.036	0.039
	[0.054]	[0.050]
Trend	0.042	0.05
	[0.012]***	[0.007]***
Observations	538780	427463
Pseudo R-Squared	0.35	0.37
-		

(The short-maturity sample is a fourth of the short-maturity observations. The long-maturity sample is the full long-maturity sample.)

## TABLE 5: Quantity Results By Maturity

An OLS regression with bank fixed effects. The dependent variable in each column is the number of loans made by a bank in a given quarter divided by the mean number of loans made by the bank in the entire sample. In the "Commit" columns, the dependent variable is based on the number of commitment loans made and, in the "Spot" columns, the dependent variable is based on the number of spot loans made. "Short Maturity" is the sample of loans with maturity less than a year and "Long Maturity" is the sample of loans with maturity of at least a year. Errors are robust and clustered by bank. Standard errors are shown in brackets with \*, \*\*, and \*\*\* indicating significance at 10%, 5%, and 1%, respectively.

	(1)	(2)	(3) (4)	ł)
	Short I	Maturity	Long Maturity	
	Commit	Spot	Commit Spo	pot
Real Federal Funds Rate	0.001	0.030	0.055 0.02	)25
	[0.006]	[0.010]***	[0.011]*** [0.014]	4]*
Baa-Aaa Spread	-0.006	0.252	0.471 0.13	39
	[0.070]	[0.104]**	[0.097]*** [0.147	47]
Yield Curve Slope	-0.023	-0.018	-0.058 0.002	)02
	[0.012]*	[0.015]	[0.017]*** [0.022	22]
Ln(Bank Assets)	0.818	0.65	0.669 0.97	<del>)</del> 74
	[0.093]***	[0.128]***	[0.112]*** [0.167]**	***
GNP Growth	-0.002	0.003	0.037 0.009	)09
	[0.005]	[0.006]	[0.007]*** [0.009	09]
Unemployment	-0.028	0.011	-0.016 -0.02	)23
	[0.017]*	[0.030]	[0.026] [0.037	37]
Trend	0.001	-0.027	0.024 -0.013	)13
	[0.003]	[0.003]***	[0.003]*** [0.005]*	5]**
Observations	11421	8887	7337 359	591
Adjusted R-Squared	0.28	0.16	0.35 0.1	.10

## TABLE 6: Logit Results By Size

A logit regression on a dummy variable indicating whether a loan is made under commitment. The regressions also include bank fixed effects. "Small Firm" is the sample of loans with loan size  $\leq$  \$100K, "Medium Firm" is the sample of loans with loan size > \$100K and  $\leq$  \$1M, and "Large Firm" is the sample of loans with loan size > \$1M. Errors are robust and clustered by bank. Standard errors are shown in brackets with \*, \*\*, and \*\*\* indicating significance at 10%, 5%, and 1%, respectively.

	(1)	(2)	(3)
	Small Firm	Medium Firm	Large Firm
Real Federal Funds Rate	-0.013	-0.025	-0.054
	[0.021]	[0.018]	[0.026]**
Short Maturity	0.525	0.25	-0.183
	[0.146]***	[0.130]*	[0.182]
Baa-Aaa Spread	-0.159	-0.05	0.023
	[0.287]	[0.278]	[0.268]
Yield Curve Slope	0.029	0.008	-0.009
	[0.030]	[0.051]	[0.033]
Collateral	0.779	0.133	-0.044
	[0.096]***	[0.078]*	[0.082]
Floating	1.635	0.857	0.381
	[0.151]***	[0.110]***	[0.163]**
Demand	0.446	0.409	-0.035
	[0.200]**	[0.153]***	[0.204]
Overnight	1.482	0.778	-0.228
	[0.372]***	[0.520]	[0.213]
Ln(Bank Assets)	-0.123	-0.415	-0.591
	[0.299]	[0.272]	[0.222]***
GNP Growth	0.007	0.013	-0.013
	[0.013]	[0.010]	[0.012]
Unemployment	-0.042	-0.007	-0.138
	[0.060]	[0.040]	[0.056]**
Trend	0.039	0.047	0.027
	[0.010]***	[0.010]***	[0.008]***
Observations	404938	366375	235394
Pseudo R-Squared	0.38	0.31	0.37

(The small firm sample is a fourth of the small firm observations.

The medium firm sample is a half of the medium firm observations.

The large firm sample is the full large firm sample.)

## TABLE 7: Quantity Results By Size

An OLS regression with bank fixed effects. The dependent variable in each column is the number of loans made by a bank in a given quarter divided by the mean number of loans made by the bank in the entire sample. In the "Commit" columns, the dependent variable is based on the number of commitment loans made and, in the "Spot" columns, the dependent variable is based on the number of spot loans made and, in the "Spot" columns, the dependent variable is based on the number of spot loans made. "Small Firm" is the sample of loans with loan size  $\leq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K, and "Large Firm" is the sample of loans with loan size  $\geq$  \$100K. Errors are robust and clustered by bank. Standard errors are shown in brackets with \*, \*\*, and \*\*\* indicating significance at 10%, 5%, and 1%, respectively.

	(1)	(2)	(3)	(4)	(5)	(6)
	Sm	all Firm	Medi	um Firm	Larg	e Firm
	Commit	Spot	Commit	Spot	Commit	Spot
Real Federal Funds Rate	0.005	0.033	0.008	0.012	0.012	0.093
	[0.007]	[0.010]***	[0.006]	[0.015]	[0.009]	[0.032]***
Baa-Aaa Spread	0.078	0.312	0.028	0.145	-0.042	-0.122
	[0.085]	[0.104]***	[0.072]	[0.139]	[0.093]	[0.223]
Yield Curve Slope	-0.031	-0.005	-0.036	0.011	-0.028	-0.004
	[0.014]**	[0.014]	[0.011]***	[0.017]	[0.013]**	[0.032]
Ln(Bank Assets)	0.84	0.703	0.726	0.804	0.392	0.744
	[0.107]***	[0.133]***	[0.086]***	[0.120]***	[0.090]***	[0.151]***
GNP Growth	0.003	0.007	0.006	0.002	-0.001	0.001
	[0.005]	[0.006]	[0.005]	[0.011]	[0.008]	[0.017]
Unemployment	-0.023	-0.028	-0.029	-0.031	-0.023	0.163
	[0.019]	[0.027]	[0.017]*	[0.036]	[0.023]	[0.074]**
Trend	0.006	-0.026	0.005	-0.023	0.006	-0.015
	[0.003]*	[0.003]***	[0.003]*	[0.004]***	[0.003]*	[0.005]***
Observations	11175	8929	9701	4111	5699	1914
Adjusted R-Squared	0.29	0.13	0.33	0.08	0.16	0.12

### TABLE 8: Logit Results By Maturity and Size

A logit regression on a dummy variable indicating whether a loan is made under commitment. The regressions also include bank fixed effects. "Short Maturity" is the sample of loans with maturity less than a year and "Long Maturity" is the sample of loans with maturity of at least a year. "Small Firm" is the sample of loans with loan size  $\leq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K. The sample of loans with loan size  $\geq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$100K, "Medium Fir

	(1)	(2)	(3)	(4)	(5)	(6)
_		all Firm		um Firm		je Firm
	Short	Long	Short	Long	Short	
	Maturity	Maturity	Maturity	Maturity		ong Maturity
Real Federal Funds Rate	-0.019	0.037	-0.041	0.035	-0.074	0.042
	[0.024]	[0.030]	[0.022]*	[0.027]	[0.025]***	[0.034]
Baa-Aaa Spread	-0.215	0.097	-0.144	0.27	-0.044	0.148
	[0.339]	[0.197]	[0.291]	[0.334]	[0.233]	[0.426]
Yield Curve Slope	0.041	-0.008	0.016	-0.048	-0.001	-0.054
	[0.031]	[0.042]	[0.050]	[0.053]	[0.027]	[0.075]
Collateral	0.814	0.72	0.28	-0.261	0.144	-0.491
	[0.085]***	[0.277]***	[0.089]***	[0.111]**	[0.084]*	[0.166]***
Floating	1.586	1.86	0.553	1.374	0.213	0.751
-	[0.182]***	[0.190]***	[0.131]***	[0.126]***	[0.183]	[0.182]***
Demand	0.52		0.375		-0.084	
	[0.214]**		[0.145]***		[0.206]	
Overnight	1.467		0.606		-0.224	
C C	[0.367]***		[0.459]		[0.199]	
Ln(Bank Assets)	-0.097	-0.231	-0.449	-0.312	-0.586	-0.525
, , , , , , , , , , , , , , , , , , ,	[0.341]	[0.170]	[0.320]	[0.199]	[0.231]**	[0.334]
GNP Growth	0.004	0.02	0.008	0.01	-0.019	0.014
	[0.015]	[0.013]	[0.012]	[0.012]	[0.013]	[0.015]
Unemployment	-0.058	0.031	-0.008	0.018	-0.139	-0.064
	[0.065]	[0.056]	[0.042]	[0.069]	[0.057]**	[0.066]
Trend	0.038	0.049	0.047	0.054	0.025	0.034
	[0.012]***	[0.007]***	[0.012]***	[0.008]***	[0.008]***	[0.010]***
	-					
Observations	347972	54665	288527	77475	184699	50555
Pseudo R-Squared	0.40	0.40	0.32	0.35	0.40	0.33

(The small firm sample is a fourth of the small firm observations.

The medium firm sample is a half of the medium firm observations.

The large firm sample is the full large firm sample.)

#### TABLE 9: Quantity Results By Maturity and Size

An OLS regression with bank fixed effects. The dependent variable in each column is the number of loans made by a bank in a given quarter divided by the mean number of loans made by the bank in the entire sample. In the "Commit" columns, the dependent variable is based on the number of commitment loans made and, in the "Spot" columns, the dependent variable is based on the number of spot loans made. "Short Maturity" is the sample of loans with maturity less than a year and "Long Maturity" is the sample of loans with maturity less than a year and "Long Maturity" is the sample of loans with maturity less than a year and "Long Maturity" is the sample of loans with maturity of at least a year. "Small Firm" is the sample of loans with loan size  $\leq$  \$100K, "Medium Firm" is the sample of loans with loan size  $\geq$  \$10K and  $\leq$  \$10, and "Large Firm" is the sample of loans with loan size  $\geq$  \$10. Errors are robust and clustered by bank. Standard errors are shown in brackets with \*, \*\*, and \*\*\* indicating significance at 10%, 5%, and 1%, respectively.

	(1)	(2)	(3)	(4)	(5)	(6)	
SHORT MATURITY	Sma	Small Firm Medium Firm L		Medium Firm		Large Firm	
	Commit	Spot	Commit	Spot	Commit	Spot	
Real Federal Funds Rate	-0.001	0.031	-0.001	0.02	0	0.125	
	[0.007]	[0.011]***	[0.006]	[0.017]	[0.010]	[0.034]***	
Baa-Aaa Spread	0.019	0.278	-0.093	0.01	-0.23	-0.245	
	[0.084]	[0.119]**	[0.074]	[0.172]	[0.106]**	[0.257]	
Yield Curve Slope	-0.028	-0.009	-0.022	0.019	-0.016	-0.008	
	[0.016]*	[0.017]	[0.011]**	[0.019]	[0.015]	[0.038]	
Ln(Bank Assets)	0.91	0.671	0.749	0.833	0.352	0.654	
	[0.103]***	[0.174]***	[0.089]***	[0.158]***	[0.104]***	[0.201]***	
GNP Growth	-0.003	0	-0.004	-0.011	-0.016	0.005	
	[0.005]	[0.007]	[0.005]	[0.012]	[0.007]**	[0.019]	
Unemployment	-0.022	-0.015	-0.032	-0.014	-0.034	0.22	
	[0.019]	[0.033]	[0.017]*	[0.043]	[0.026]	[0.092]**	
Trend	0.002	-0.029	-0.001	-0.026	-0.004	-0.013	
	[0.003]	[0.004]***	[0.003]	[0.005]***	[0.003]	[0.006]**	
Observations	10995	7721	9555	3147	4738	1549	
Adjusted R-Squared	0.28	0.15	0.27	0.09	0.07	0.12	

	(7)	(8)	(9)	(10)	(11)	(12)
LONG MATURITY	Sma	Small Firm		ım Firm	Large	Firm
	Commit	Spot	Commit	Spot	Commit	Spot
Real Federal Funds Rate	0.034	0.009	0.046	-0.037	0.045	-0.031
	[0.013]**	[0.019]	[0.012]***	[0.052]	[0.015]***	[0.038]
Baa-Aaa Spread	0.566	0.323	0.572	-0.67	0.727	0.864
	[0.126]***	[0.203]	[0.114]***	[0.503]	[0.154]***	[0.925]
Yield Curve Slope	-0.048	0	-0.071	-0.08	-0.092	0.031
	[0.025]*	[0.029]	[0.018]***	[0.055]	[0.025]***	[0.188]
Ln(Bank Assets)	0.537	0.631	0.527	0.843	0.477	1.747
	[0.144]***	[0.259]**	[0.132]***	[0.363]**	[0.151]***	[0.986]
GNP Growth	0.035	0.009	0.042	-0.021	0.045	-0.005
	[0.008]***	[0.013]	[0.008]***	[0.036]	[0.011]***	[0.057]
Unemployment	-0.009	-0.105	-0.029	0.195	-0.066	-0.077
	[0.040]	[0.053]*	[0.033]	[0.134]	[0.046]	[0.073]
Trend	0.028	-0.005	0.029	-0.009	0.028	-0.017
	[0.004]***	[0.008]	[0.004]***	[0.006]	[0.005]***	[0.044]
Observations	5217	2163	4403	590	1948	121
Adjusted R-Squared	0.37	0.07	0.48	0.03	0.48	0.52

### TABLE 10: Logit and Quantity Results by Bank Size, Maturity, and Firm Size

The coefficients on the real federal funds rate for both the logit and quantity regressions by firm size, maturity, and bank size. The logit regressions are on a dummy variable indicating whether a loan is made under commitment. The quantity regressions are OLS regressions in which the dependent variable in each column is the number of loans made by a bank in a given quarter divided by the mean number of loans made by the bank in the entire sample. In the "Commit" columns, the dependent variable is based on the number of commitment loans made and, in the "Spot" columns, the dependent variable is based on the number of spot loans made. The regressions all include bank fixed effects. The number of observations are shown under *Obs*.

"Short Maturity" is the sample of loans with maturity less than a year and "Long Maturity" is the sample of loans with maturity of at least a year. "Small Firm" is the sample of loans with loan size  $\leq$  \$100K, "Medium Firm" is the sample of loans with loan size > \$100K and  $\leq$  \$11M, and "Large Firm" is the sample of loans with loan size > \$100K. "Small Bank" is the sample of loans made by a bank with total assets less than or equal to \$10 billion and "Large Bank" is the sample of loans made by a bank with total assets of at least \$10 billion. All nominal values are in 2006 dollars. Errors are robust and clustered by bank. The \*, \*\*, and \*\*\* indicate significance at 10%, 5%, and 1%, respectively.

	Lo	git		Qu	antity	
			Com	mit	Spo	ot
	Coeff.	Obs.	Coeff.	Obs.	Coeff.	Obs.
Small Bank	0.013	802690	0.011	7323	0.015	6130
Short maturity	0.008	695519	0.006	7192	0.018	5254
Small firm	0.010	475973	0.009	4707	0.033 **	2926
Medium firm	0.015	179068	0.007	3185	-0.015	416
Large firm	-0.036	37041	0.005	333	0.137	103
Long maturity	0.052 **	106770	0.041 ***	3402	0.021	1421
Small firm	0.042	67951	0.038 *	1066	0.005	408
Medium firm	0.061 **	32235	0.033 ***	590	0.013	23
Large firm	0.097 ***	6189	0.002 **	44	n/a	n/a
Large Bank	-0.041 *	1784717	-0.003	6546	0.057 ***	3917
Short maturity	-0.053 **	1464265	-0.01	4229	0.064 ***	3633
Small firm	-0.039	920283	-0.014	6288	0.046 ***	4795
Medium firm	-0.071 **	396550	-0.009	6370	0.035 *	2731
Large firm	-0.080 **	146637	0.001	4405	0.132 ***	1446
Long maturity	0.022	320201	0.054 ***	3935	0.046 *	2170
Small firm	0.040	151654	0.028	4151	0.017	1755
Medium firm	0.010	123751	0.043 ***	3813	-0.044	567
Large firm	0.020	43702	0.046 ***	1904	-0.031	121

## **Working Paper Series**

A series of research studies on regional economic issues relating to the Seventh Federal Reserve District, and on financial and economic topics.

Standing Facilities and Interbank Borrowing: Evidence from the Federal Reserve's New Discount Window <i>Craig Furfine</i>	WP-04-01
Netting, Financial Contracts, and Banks: The Economic Implications William J. Bergman, Robert R. Bliss, Christian A. Johnson and George G. Kaufman	WP-04-02
Real Effects of Bank Competition Nicola Cetorelli	WP-04-03
Finance as a Barrier To Entry: Bank Competition and Industry Structure in Local U.S. Markets? Nicola Cetorelli and Philip E. Strahan	WP-04-04
The Dynamics of Work and Debt Jeffrey R. Campbell and Zvi Hercowitz	WP-04-05
Fiscal Policy in the Aftermath of 9/11 Jonas Fisher and Martin Eichenbaum	WP-04-06
Merger Momentum and Investor Sentiment: The Stock Market Reaction To Merger Announcements <i>Richard J. Rosen</i>	WP-04-07
Earnings Inequality and the Business Cycle Gadi Barlevy and Daniel Tsiddon	WP-04-08
Platform Competition in Two-Sided Markets: The Case of Payment Networks Sujit Chakravorti and Roberto Roson	WP-04-09
Nominal Debt as a Burden on Monetary Policy Javier Díaz-Giménez, Giorgia Giovannetti, Ramon Marimon, and Pedro Teles	WP-04-10
On the Timing of Innovation in Stochastic Schumpeterian Growth Models <i>Gadi Barlevy</i>	WP-04-11
Policy Externalities: How US Antidumping Affects Japanese Exports to the EU Chad P. Bown and Meredith A. Crowley	WP-04-12
Sibling Similarities, Differences and Economic Inequality Bhashkar Mazumder	WP-04-13
Determinants of Business Cycle Comovement: A Robust Analysis Marianne Baxter and Michael A. Kouparitsas	WP-04-14
The Occupational Assimilation of Hispanics in the U.S.: Evidence from Panel Data <i>Maude Toussaint-Comeau</i>	WP-04-15

Reading, Writing, and Raisinets <sup>1</sup> : Are School Finances Contributing to Children's Obesity? <i>Patricia M. Anderson and Kristin F. Butcher</i>	WP-04-16
Learning by Observing: Information Spillovers in the Execution and Valuation of Commercial Bank M&As <i>Gayle DeLong and Robert DeYoung</i>	WP-04-17
Prospects for Immigrant-Native Wealth Assimilation: Evidence from Financial Market Participation Una Okonkwo Osili and Anna Paulson	WP-04-18
Individuals and Institutions: Evidence from International Migrants in the U.S. <i>Una Okonkwo Osili and Anna Paulson</i>	WP-04-19
Are Technology Improvements Contractionary? Susanto Basu, John Fernald and Miles Kimball	WP-04-20
The Minimum Wage, Restaurant Prices and Labor Market Structure Daniel Aaronson, Eric French and James MacDonald	WP-04-21
Betcha can't acquire just one: merger programs and compensation <i>Richard J. Rosen</i>	WP-04-22
Not Working: Demographic Changes, Policy Changes, and the Distribution of Weeks (Not) Worked <i>Lisa Barrow and Kristin F. Butcher</i>	WP-04-23
The Role of Collateralized Household Debt in Macroeconomic Stabilization <i>Jeffrey R. Campbell and Zvi Hercowitz</i>	WP-04-24
Advertising and Pricing at Multiple-Output Firms: Evidence from U.S. Thrift Institutions <i>Robert DeYoung and Evren Örs</i>	WP-04-25
Monetary Policy with State Contingent Interest Rates Bernardino Adão, Isabel Correia and Pedro Teles	WP-04-26
Comparing location decisions of domestic and foreign auto supplier plants Thomas Klier, Paul Ma and Daniel P. McMillen	WP-04-27
China's export growth and US trade policy Chad P. Bown and Meredith A. Crowley	WP-04-28
Where do manufacturing firms locate their Headquarters? J. Vernon Henderson and Yukako Ono	WP-04-29
Monetary Policy with Single Instrument Feedback Rules Bernardino Adão, Isabel Correia and Pedro Teles	WP-04-30

Firm-Specific Capital, Nominal Rigidities and the Business Cycle David Altig, Lawrence J. Christiano, Martin Eichenbaum and Jesper Linde	WP-05-01
Do Returns to Schooling Differ by Race and Ethnicity? Lisa Barrow and Cecilia Elena Rouse	WP-05-02
Derivatives and Systemic Risk: Netting, Collateral, and Closeout Robert R. Bliss and George G. Kaufman	WP-05-03
Risk Overhang and Loan Portfolio Decisions Robert DeYoung, Anne Gron and Andrew Winton	WP-05-04
Characterizations in a random record model with a non-identically distributed initial record <i>Gadi Barlevy and H. N. Nagaraja</i>	WP-05-05
Price discovery in a market under stress: the U.S. Treasury market in fall 1998 <i>Craig H. Furfine and Eli M. Remolona</i>	WP-05-06
Politics and Efficiency of Separating Capital and Ordinary Government Budgets Marco Bassetto with Thomas J. Sargent	WP-05-07
Rigid Prices: Evidence from U.S. Scanner Data Jeffrey R. Campbell and Benjamin Eden	WP-05-08
Entrepreneurship, Frictions, and Wealth Marco Cagetti and Mariacristina De Nardi	WP-05-09
Wealth inequality: data and models Marco Cagetti and Mariacristina De Nardi	WP-05-10
What Determines Bilateral Trade Flows? Marianne Baxter and Michael A. Kouparitsas	WP-05-11
Intergenerational Economic Mobility in the U.S., 1940 to 2000 Daniel Aaronson and Bhashkar Mazumder	WP-05-12
Differential Mortality, Uncertain Medical Expenses, and the Saving of Elderly Singles Mariacristina De Nardi, Eric French, and John Bailey Jones	WP-05-13
Fixed Term Employment Contracts in an Equilibrium Search Model Fernando Alvarez and Marcelo Veracierto	WP-05-14
Causality, Causality, Causality: The View of Education Inputs and Outputs from Economics Lisa Barrow and Cecilia Elena Rouse	WP-05-15

Competition in Large Markets Jeffrey R. Campbell	WP-05-16
Why Do Firms Go Public? Evidence from the Banking Industry Richard J. Rosen, Scott B. Smart and Chad J. Zutter	WP-05-17
Clustering of Auto Supplier Plants in the U.S.: GMM Spatial Logit for Large Samples <i>Thomas Klier and Daniel P. McMillen</i>	WP-05-18
Why are Immigrants' Incarceration Rates So Low? Evidence on Selective Immigration, Deterrence, and Deportation <i>Kristin F. Butcher and Anne Morrison Piehl</i>	WP-05-19
Constructing the Chicago Fed Income Based Economic Index – Consumer Price Index: Inflation Experiences by Demographic Group: 1983-2005 Leslie McGranahan and Anna Paulson	WP-05-20
Universal Access, Cost Recovery, and Payment Services Sujit Chakravorti, Jeffery W. Gunther, and Robert R. Moore	WP-05-21
Supplier Switching and Outsourcing Yukako Ono and Victor Stango	WP-05-22
Do Enclaves Matter in Immigrants' Self-Employment Decision? Maude Toussaint-Comeau	WP-05-23
The Changing Pattern of Wage Growth for Low Skilled Workers Eric French, Bhashkar Mazumder and Christopher Taber	WP-05-24
U.S. Corporate and Bank Insolvency Regimes: An Economic Comparison and Evaluation <i>Robert R. Bliss and George G. Kaufman</i>	WP-06-01
Redistribution, Taxes, and the Median Voter Marco Bassetto and Jess Benhabib	WP-06-02
Identification of Search Models with Initial Condition Problems Gadi Barlevy and H. N. Nagaraja	WP-06-03
Tax Riots Marco Bassetto and Christopher Phelan	WP-06-04
The Tradeoff between Mortgage Prepayments and Tax-Deferred Retirement Savings Gene Amromin, Jennifer Huang, and Clemens Sialm	WP-06-05
Why are safeguards needed in a trade agreement? Meredith A. Crowley	WP-06-06

Taxation, Entrepreneurship, and Wealth Marco Cagetti and Mariacristina De Nardi	WP-06-07
A New Social Compact: How University Engagement Can Fuel Innovation Laura Melle, Larry Isaak, and Richard Mattoon	WP-06-08
Mergers and Risk Craig H. Furfine and Richard J. Rosen	WP-06-09
Two Flaws in Business Cycle Accounting Lawrence J. Christiano and Joshua M. Davis	WP-06-10
Do Consumers Choose the Right Credit Contracts? Sumit Agarwal, Souphala Chomsisengphet, Chunlin Liu, and Nicholas S. Souleles	WP-06-11
Chronicles of a Deflation Unforetold François R. Velde	WP-06-12
Female Offenders Use of Social Welfare Programs Before and After Jail and Prison: Does Prison Cause Welfare Dependency? <i>Kristin F. Butcher and Robert J. LaLonde</i>	WP-06-13
Eat or Be Eaten: A Theory of Mergers and Firm Size Gary Gorton, Matthias Kahl, and Richard Rosen	WP-06-14
Do Bonds Span Volatility Risk in the U.S. Treasury Market? A Specification Test for Affine Term Structure Models <i>Torben G. Andersen and Luca Benzoni</i>	WP-06-15
Transforming Payment Choices by Doubling Fees on the Illinois Tollway Gene Amromin, Carrie Jankowski, and Richard D. Porter	WP-06-16
How Did the 2003 Dividend Tax Cut Affect Stock Prices? Gene Amromin, Paul Harrison, and Steven Sharpe	WP-06-17
Will Writing and Bequest Motives: Early 20th Century Irish Evidence Leslie McGranahan	WP-06-18
How Professional Forecasters View Shocks to GDP Spencer D. Krane	WP-06-19
Evolving Agglomeration in the U.S. auto supplier industry Thomas Klier and Daniel P. McMillen	WP-06-20
Mortality, Mass-Layoffs, and Career Outcomes: An Analysis using Administrative Data Daniel Sullivan and Till von Wachter	WP-06-21

The Agreement on Subsidies and Countervailing Measures: Tying One's Hand through the WTO. <i>Meredith A. Crowley</i>	WP-06-22
How Did Schooling Laws Improve Long-Term Health and Lower Mortality? Bhashkar Mazumder	WP-06-23
Manufacturing Plants' Use of Temporary Workers: An Analysis Using Census Micro Data Yukako Ono and Daniel Sullivan	WP-06-24
What Can We Learn about Financial Access from U.S. Immigrants? Una Okonkwo Osili and Anna Paulson	WP-06-25
Bank Imputed Interest Rates: Unbiased Estimates of Offered Rates? Evren Ors and Tara Rice	WP-06-26
Welfare Implications of the Transition to High Household Debt <i>Jeffrey R. Campbell and Zvi Hercowitz</i>	WP-06-27
Last-In First-Out Oligopoly Dynamics Jaap H. Abbring and Jeffrey R. Campbell	WP-06-28
Oligopoly Dynamics with Barriers to Entry Jaap H. Abbring and Jeffrey R. Campbell	WP-06-29
Risk Taking and the Quality of Informal Insurance: Gambling and Remittances in Thailand Douglas L. Miller and Anna L. Paulson	WP-07-01
Fast Micro and Slow Macro: Can Aggregation Explain the Persistence of Inflation? Filippo Altissimo, Benoît Mojon, and Paolo Zaffaroni	WP-07-02
Assessing a Decade of Interstate Bank Branching Christian Johnson and Tara Rice	WP-07-03
Debit Card and Cash Usage: A Cross-Country Analysis Gene Amromin and Sujit Chakravorti	WP-07-04
The Age of Reason: Financial Decisions Over the Lifecycle Sumit Agarwal, John C. Driscoll, Xavier Gabaix, and David Laibson	WP-07-05
Information Acquisition in Financial Markets: a Correction Gadi Barlevy and Pietro Veronesi	WP-07-06
Monetary Policy, Output Composition and the Great Moderation Benoît Mojon	WP-07-07
Estate Taxation, Entrepreneurship, and Wealth Marco Cagetti and Mariacristina De Nardi	WP-07-08

Conflict of Interest and Certification in the U.S. IPO Market Luca Benzoni and Carola Schenone	WP-07-09
The Reaction of Consumer Spending and Debt to Tax Rebates – Evidence from Consumer Credit Data Sumit Agarwal, Chunlin Liu, and Nicholas S. Souleles	WP-07-10
Portfolio Choice over the Life-Cycle when the Stock and Labor Markets are Cointegrated Luca Benzoni, Pierre Collin-Dufresne, and Robert S. Goldstein	WP-07-11
Nonparametric Analysis of Intergenerational Income Mobility with Application to the United States Debopam Bhattacharya and Bhashkar Mazumder	WP-07-12
How the Credit Channel Works: Differentiating the Bank Lending Channel and the Balance Sheet Channel Lamont K. Black and Richard J. Rosen	WP-07-13